cambridge.org/pab

# Article

**Cite this article:** Phillipi DB, Czekanski-Moir J, Ivany LC (2024). Global Phanerozoic biodiversity—can variation be explained by spatial sampling intensity? *Paleobiology* **50**, 661–672. https://doi.org/10.1017/pab.2024.26

Received: 14 September 2023 Revised: 21 July 2024 Accepted: 5 August 2024

Corresponding authors: Daniel B. Phillipi; Email: daniel.phillipi.137@gmail.com; Linda C. Ivany; Email: lcivany@syr.edu

© The Author(s), 2025. Published by Cambridge University Press on behalf of Paleontological Society. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited.



# Global Phanerozoic biodiversity—can variation be explained by spatial sampling intensity?

Daniel B. Phillipi<sup>1</sup> (1), Jesse Czekanski-Moir<sup>2</sup> (1) and Linda C. Ivany<sup>1</sup> (1)

<sup>1</sup>Department of Earth and Environmental Sciences, Syracuse University, Syracuse, New York 13244, U.S.A. <sup>2</sup>Department of Environmental Biology, SUNY College of Environmental Science and Forestry, Syracuse, New York 13210, U.S.A.

## **Non-technical Summary**

The number of species identified in the fossil record within any given geological time period must partly be explained by both the total number of fossil specimens sampled from that interval and the geographic spread of those samples. The influence of numerical sampling intensity has been well studied, but the effects of geographic variance in sampling are comparatively unknown. To investigate this question, we repeatedly resample a dataset of modern global marine organisms (from the Ocean Biodiversity Information System) using spatial sampling parameters determined by the real geographic spread of fossil organisms as found in the Paleobiology Database. Our findings show that a significant proportion of the variance in fossil diversity through time can be attributed only to changes in the numbers and locations of fossils sampled. This is consistent with the claim that the spatial structure of the fossil record and how it is sampled largely determine the diversity history drawn from it and with the possibility that global diversity has been relatively constant over time.

## Abstract

Variation in observed global generic richness over the Phanerozoic must be partly explained by changes in the numbers of fossils and their geographic spread over time. The influence of sampling intensity (i.e., the number of samples) has been well addressed, but the extent to which the geographic distribution of samples might influence recovered biodiversity is comparatively unknown. To investigate this question, we create models of genus richness through time by resampling the same occurrence dataset of modern global biodiversity using spatially explicit sampling intensities defined by the paleo-coordinates of fossil occurrences from successive time intervals. Our steady-state null model explains about half of observed change in uncorrected fossil diversity and a quarter of variation in sampling-standardized diversity estimates. The inclusion in linear models of two additional explanatory variables associated with the spatial array of fossil data (absolute latitudinal range of occurrences, percentage of occurrences from shallow environments) and a Cenozoic step increases the accuracy of steady-state models, accounting for 67% of variation in sampling-standardized estimates and more than one-third of the variation in first differences. Our results make clear that the spatial distribution of samples is at least as important as numerical sampling intensity in determining the trajectory of recovered fossil biodiversity through time and caution against the overinterpretation of both the variation and the trend that emerge from analyses of global Phanerozoic diversity.

# Introduction

The incomplete and unevenly sampled nature of the fossil record was recognized as an obstacle to accurate reconstructions of trends in global Phanerozoic biodiversity as early as the midnineteenth century (Phillips 1860). Temporal unevenness in the numbers of recovered fossils (numerical sampling bias) inflates observed richness during time intervals from which more fossils have been reported. Traditionally employed by paleoecologists in small-scale regional studies, occurrence-based subsampling was adopted by the architects of the Paleobiology Database (PBDB) in an effort to recover a more accurate picture of changes in global richness over the Phanerozoic (Alroy et al. 2001, 2008). While the details differ across approaches (Alroy et al. 2001, 2008; Alroy 2010; Chao and Jost 2012; Close et al. 2018), subsampling yields increasingly accurate estimates of the temporal changes in both local and global biodiversity over the Phanerozoic.

Sampling bias, however, is not only numerical in origin—the spatial structure of fossil data may be critical as well (Close et al. 2020b; Antell et al. 2024; Fig. 1). Spatial biases in the way the fossil record is sampled differ fundamentally from numerical biases in that the geographic or environmental distribution of data is the primary concern. Biodiversity estimates today, as in the past, are directly influenced by the uneven distribution of taxa over the globe (Buffon





Figure 1. Comparison of Paleobiology Database (PBDB) sampling in two distinct Paleozoic time bins, illustrating the wide variation in geographic sampling completeness within the PBDB. While it is true that the ca. 447 Ma time bin has more occurrences, those occurrences are also spread out across latitude and multiple ocean basins. On the other hand, the ca. 303 Ma time bin has occurrences concentrated in only one ocean basin and within a relatively small latitudinal band.

1774; von Humboldt and Bonpland 1807). The nature of the geological record ensures that the sediments that might record this heterogeneity are neither deposited nor preserved with equal frequency over the Earth's surface. A first-order pattern in fossil diversity studies arises from the tendency of time intervals preserving more area and volume of sedimentary rock to yield more samples and hence more recovered taxa (Raup 1976; Crampton et al. 2003; Smith and McGowan 2007; Ye and Peters 2023). Fossil preservation is also nonrandom across paleoenvironments, and different environments are not consistently represented in the record (Schopf 1978; Peters 2007; Patzkowsky and Holland 2012; Shaw et al. 2020). Furthermore, beyond the spatial inequities of sampling dictated by geological phenomena, a disproportionate percentage of fossil occurrences come from North America and Europe (Alroy et al. 2001; McGowan and Smith 2008; Ye and Peters 2023) due largely to socioeconomic, political, and historical circumstances, with implications for the accuracy of biodiversity reconstructions (Raja et al. 2022).

Previous studies have acknowledged and, in some cases, attempted to address variation in richness resulting from inconsistent spatial sampling by accounting for changes in outcrop area with time (Raup 1976; Smith and Gowan 2007; Wall et al. 2009). While this relates to the broader issue of spatial sampling bias, outcrop area alone is a poor proxy for spatial variation in sampling, because spatial biases originate from both the quantity *and* the spatial distribution of data. More recently, Close et al. (2020a,b) used minimum spanning trees (MST) to approximate the spatial coverage of fossil occurrence data and found a strong correlation with numerical sampling-standardized diversity estimates. Reconstruction of biogeographic patterns even as fundamental as the latitudinal diversity gradient (LDG) can thus be complicated by changes in the spatial extent of fossil sampling (Jones et al. 2021; Antell et al. 2024).

Here, we assess the impact of spatial sampling biases on our ability to reconstruct changes in global marine fossil richness over the Phanerozoic in a different way. Instead of subsampling fossil-record data from the PBDB through time, we repeatedly resample the same single dataset, in this case, modern marine skeletonized invertebrate biodiversity retrieved from the Ocean Biodiversity Information System (OBIS), using spatial sampling parameters defined by occurrences in the PBDB from different time intervals through the Phanerozoic. By holding real global richness static and subsampling that same dataset in different spatially explicit ways, we create a model for the Phanerozoic wherein all changes in recovered biodiversity are due *only* to changes in the spatial distribution of available occurrences through time. Comparing this purely spatially driven model with recovered fossil diversity over the Phanerozoic reveals the extent to which estimated changes in fossil biodiversity are driven by spatial sampling factors alone.

## **Methods**

#### Datasets

Fossil data spanning the Phanerozoic were downloaded from the PBDB (here on 24 September 2021) and include global marine skeletonized macroinvertebrate occurrences belonging to six major phyla (Arthropoda, Brachiopoda, Cnidaria, Echinodermata, Mollusca, and Porifera), excluding occurrences with uncertain generic assignments and those with an age uncertainty greater than the longest Phanerozoic stage (~18 Myr). Primary references from which the data are drawn include Aberhan (1992), Cooper and Grant (1977), Foster et al. (2020), Holland and Patzkowsky (2007), Fürsich (1999), Jung (1989), Manivit et al. (1990), Reboulet (1996), Reed (1944), and some 13,000 others. The PBDB dataset is analyzed and cleaned manually to remove nonmarine groups, out-of-date or poor taxonomy, and obvious errors. The final dataset contains 23,656 genera spread across 48 subequal ~11 Myr time bins that do not cross stage boundaries (following Alroy et al. 2008). Each time bin contains between 3259 and 46,310 occurrences and 353 and 2882 genera. The full dataset contains 590,144 occurrences.

Data on modern marine biodiversity are drawn from the OBIS (here on 26 September 2022) using the robis package (Provoost et al. 2022). The database is searched for generic occurrences belonging to all extant taxonomic orders appearing in the PBDB dataset defined previously. Any OBIS occurrences marked as fossil are removed. The OBIS dataset is transformed into a geographic shapefile (Pebesma et al. 2018) and clipped to exclusively shallow-marine environments based on modern continental shelf geometries. Decapods are removed because they comprise an artificially high proportion of the modern diversity dataset. Finally, after OBIS occurrences have been grouped into equal-area grid cells (see below), cells with fewer than 1000 occurrences are removed, as these cells often represent aberrations in the dataset (e.g., latitude and longitude entered in reverse order) or contain very little shallow area (e.g., cells that mostly contain the deep ocean but may intersect a small part of the continental margins). The final OBIS dataset contains 3336 genera across 47 extant orders, for a total of 3,438,977 occurrences.

# Quantification of Fossil Spatial Sampling through Time

To describe the geographic distribution of sampling in the PBDB and OBIS datasets, occurrences in both are first geographically binned into 812 equal-area grid cells covering the Earth surface using the R package dggridR (Barnes and Sahr 2017). Each grid cell has an area of approximately 630,000 km<sup>2</sup>, and the mean distance between cell centroids is 855 km. The number of grid cells containing fossil data are counted for each of 48 time bins, and cells are grouped into five latitudinal bands (90–51°S, 50–23.4°S, 23.4°S–23.4°N, 23.4–50°N, and 51–90°N). Counts of cells and their geographic positions form the basis for the selection of modern OBIS cells for resampling. Fossil occurrence and cell MST distances are also calculated using the igraph (Csardi and Nepusz 2006) and sf (Pebesma et al. 2018) R packages. PBDB dataset variables, including the number of occurrences, percent shallow and percent tropical occurrences, and number of publications, are also recorded for each time bin.

### Resampling the OBIS Dataset

First, a smaller subset (20%) of the OBIS dataset is randomly selected. This is done to increase processing speed and allow for the quantification of error based on random resampling. Modern cells containing OBIS data are selected to best match the latitudinal and longitudinal distributions of PBDB-occupied cells for each Phanerozoic time bin. To do this, modern cells are first assigned to the appropriate latitudinal band and then shifted eastward until the longitudinal distribution of modern cells within a latitudinal band completely covers the longitudinal distribution of PBDB cells in a given time bin (Fig. 2). For example, if paleo-occurrences in a time bin occupy three cells in the high north latitudinal bin, a set of equivalent OBIS cells will either be selected at those same longitude coordinates (if they contain data) or shifted together until cells can be sampled from those coordinates. In this way, the resampling procedure preserves absolute latitude (important for constraining the LDG) and captures the longitudinal spread of samples (which positively correlates with diversity at a given latitude today; Rosenzweig 1995; Dallas et al. 2024; He et al. 2024). Given that there are many more modern cells containing data than there are paleo-cells in any time bin, longitudinal shifting is usually minimal and is most prevalent in the earliest stages of the Phanerozoic. There are always enough modern cells to cover the paleo-distribution with error less than ±25° of longitude.

After a suite of potential representative modern cells are selected, a subset is randomly chosen such that the number of modern cells is equivalent to the number of paleo-cells within a given latitudinal band and their longitudinal distributions match. After the cells have been selected, every OBIS occurrence from those cells is grouped into a dataset, and the number of unique genera is counted. In this way, the resampling protocol assumes complete sampling (or at least as complete as the OBIS dataset) of every PBDB-defined cell that contains data, which is an unrealistically optimistic scenario for fossil sampling but a worst-case scenario for our spatial model. To quantify error, the resampling process is run with a different random subset of the OBIS dataset 25 times.

Resampled OBIS diversity metrics are then compared with the actual fossil record of diversity, both raw (sampled in bin [SIB]) and sampling standardized (shareholder quorum subsampling [SQS]; Alroy 2010) as computed from the fossil data with the R package divDyn (Kocsis et al. 2019), using linear regression analysis. Fossil diversity is also compared using linear regression to a variety of metrics computed from the PBDB dataset that broadly fall into three categories: sampling intensity factors (number of occurrences, number of publications, percentage of occurrences composed of any aragonite), spatial factors (number of cells,



**Figure 2.** Schematic showing an overview of the geographic subsampling process, which imposes the sampling "design" the fossil record has "given" us from each of 48 ~11 Ma time bins onto the Ocean Biodiversity Information System (OBIS) dataset of modern marine diversity for the same major groups on continental shelves. **A**, Geography and sampling intensity of a given latitudinal band from the Paleobiology Database (PBDB), with the pooled number of genera (top number) and occurrences (bottom number) within occupied equal-area cells. **B**, Geography of available modern data in the same latitudinal band, which are shifted together to the east (with potential wraparound) until their east–west distribution roughly matches the longitudinal distributions of PBDB occurrence data. **C**, Probability of sampling from each cell within a geographic cluster (e.g., if there are five bins that approximately match the position of the PBDB data, each of those has a probability of 0.2 of being sampled in a given run). **D**, Modern OBIS cells with approximately geography are subsampled so as to match the sampling intensity (no. of occurrences) of corresponding cells in the PBDB dataset, and the resulting number of genera is recorded for each. This process is repeated 25 times using a different random subsample of OBIS each time. See the text for more details.

summed MST distance, latitudinal range of occurrences), and environmental factors (percentage of occurrences from tropical, reefal, shallow, and carbonate environments). Models are created using each PBDB metric alone, in combination with each other, and in combination with the OBIS resampled diversity estimate. Bin-to-bin first differences are also calculated for all diversity and additional dataset variables, and these are similarly subjected to linear regression analysis. Adjusted  $R^2$  (adj.  $R^2$ ) is calculated with the R statistical programming language and differs from  $R^2$ in that the number of independent regressors (i.e., the number of independent variables) is considered in the calculation. It always produces a value less than or equal to  $R^2$ , and the explicit consideration of the number of variables being fit to a model helps reduce the effects of overfitting. Akaike information criterion (AIC) (Bozdogan 1987) model ranking scores are additionally used to compare the relative quality of statistical models—lower numbers represent higher quality, and an absolute difference between two represents a statistically significant difference in quality. AIC scores estimate model quality in a relative manner, so the values themselves are only relevant for models derived from the same dataset.

# Results

## Spatial Sampling in the Fossil and Modern Datasets

The number of equal-area grid cells occupied by marine-shelf fossil data per time interval varies from 53 to 195 (out of 812 total over the globe) but shows no trend through time (adj.  $R^2 =$ -0.019, p = 0.730; Fig. 3). As is true of occurrences (Alroy 2010), the latitudinal distribution of occupied cells is weighted toward the tropics in Paleozoic time bins and the northern midlatitudes during the Meso-Cenozoic. In comparison to the fossil record, modern marine shelves are both more completely and more evenly sampled, with 303 cells sampled and a latitudinal distribution of occupied cells that closely approximates that of all cells available to sample.

# Resampled Modern Biodiversity versus PBDB Richness Estimates

Spatially resampled modern diversity varies across simulated time bins, with the greatest peaks in the Late Ordovician, middle Permian, and late Cenozoic (Fig. 4). All measures of genus richness are log transformed to reduce the influence of artificially high outliers, for example, the last Cenozoic time bin. Table 1 shows the results of linear correlations between resampled OBIS diversity, recovered fossil diversity, and the additional sample descriptors derived from the PBDB dataset. Without any additional adjustments, spatially resampled modern diversity correlates significantly with fossil diversity derived from both raw uncorrected, SIB diversity ( $R^2 = 0.408$ , p < 0.001) and sampling-standardized diversity estimates from SQS (Alroy 2010) ( $R^2 = 0.247$ , p < 0.001) (Table 1). Furthermore, the first differences in resampled modern diversity correlate with the first differences in raw and standardized fossil diversity ( $R^2 = 0.305$ , p < 0.001 for SIB,  $R^2 = 0.184$ , p = 0.001 for SQS) (Fig. 4B,C). As anticipated, bin-to-bin change in the number of fossil occurrences correlates strongly with SIB fossil diversity ( $R^2 = 0.642$ , p < 0.001 for value,  $R^2 = 0.647$ , p < 0.001 for first differences), but the correlation of occurrences with SQS diversity and first differences is low ( $R^2 = 0.183$ , p = 0.001) and nonexistent ( $R^2 = 0.018$ , p = 0.180), respectively, indicating that SQS standardization effectively removes the influence of variation in the number of samples. All SQS simulations were performed with a quorum value of 0.4.

## Comparison with a Spatially Randomized Dataset

By randomizing the taxonomic column in the OBIS dataset, we can create a modern biodiversity dataset that has no relevant spatial relationships (a taxon is equally likely to appear anywhere, with probability dependent only upon its number of occurrences), and this dataset can be used to test the importance of geography in our model results. While the spatially randomized dataset is still a significant predictor of the SQS value ( $R^2 = 0.204$ , p = 0.001), the correlation between first differences is eliminated ( $R^2 = 0.023$ , p = 0.156) (Table 1). A similar pattern holds when only brachiopods are analyzed; the spatial model is highly predictive of brachiopod-only SQS value ( $R^2 = 0.390$ , p > 0.001), but there is no significant relationship with first differences ( $R^2 = 0.054$ , p = 0.062).

#### Model Residuals

Spatially resampled model residuals were compared against PBDB dataset variables with potential explanatory power for Phanerozoic diversity; these include measures of sampling intensity (number of occurrences, publications), geography (latitudinal range of occurrences, number of equal-area cells sampled), and paleoenvironment (percentage of shallow or tropical occurrences), as well as temporal



Figure 3. Spatial sampling through the Phanerozoic compared with modern data. The number of equal-area grid cells sampled across time within the coarse latitudinal bins used in this study, with cells sampled in the Ocean Biodiversity Information System (OBIS) dataset shown at the farthest right. Spatial sampling through time does not show a linear increase or decrease, but latitudinal sampling shifts northward, tracking the location of North America and Europe over the Phanerozoic; OBIS latitudinal sampling is more complete for every latitudinal band than every Paleobiology Database (PBDB) time bin. CM, Cambrian; O, Ordovician; S, Silurian; D, Devonian; C, Carboniferous; P, Permian; Tr, Triassic; J, Jurassic; K, Cretaceous; Pg, Paleogene; Ng, Neogene.



**Figure 4.** Comparison between three iterations of the resampled modern Ocean Biodiversity Information System (OBIS) model, which estimates expected fossil diversity based on only the spatial distribution of fossil data, and sampling-standardized shareholder quorum subsampling (SQS) richness of fossil marine invertebrate genera drawn from the Paleobiology Database (PBDB) (dashed black line). The raw OBIS model (blue), based only on resampled OBIS data, is adjusted to reflect the step increase into the Cenozoic (purple). The four-variable OBIS model (in red) further incorporates two additional variables: percent of shallow occurrences and latitudinal range of occurrences. **A,** Secular trends in both resampled model and SQS diversity estimates, with two-sigma error envelopes derived from replicate resampling. **B,** Correlation between the values of the four-variable OBIS model (best fit of all OBIS models based on adj. *R*<sup>2</sup> and Akaike information riterion [AIC] values) and fossil SQS. **C,** Correlation between the time bin to time bin first differences of those estimates, showing that the model generally predicts the magnitude and direction of change in fossil SQS diversity through time. The model explains about 67% of variance in the value of fossil SQS and 38% of the variance in bin-to-bin change. CM, Cambrian; O, Ordovician; S, Silurian; D, Devonian; C, Carboniferous; P, Permian; Tr, Triassic; J, Jurassic; K, Cretaceous; Pg, Paleogene; Ng, Neogene.

variables. Model residuals correlate with age in that the model substantially underestimates Cenozoic diversity. A model with a stepwise increase in diversity at the start of the Cenozoic improves the correlation and is clearly favored over one with a continuous increase in diversity through the Phanerozoic (adj.  $R^2 = 0.528$  vs. 0.428, respectively; Table 1). When the Cenozoic age factor is included in the model, the correlation between resampled modern diversity and fossil SQS diversity increases substantially for value but changes little for first differences. After incorporation of the stepwise temporal variable, the remaining substantial offsets between the resampled modern diversity model and fossil SQS diversity can be accounted for with only two additional PBDB dataset parameters—percentage of shallow occurrences and absolute latitudinal range of occurrences (i.e., the highest latitude minus the lowest latitude, 0 to 180). The inclusion of these additional explanatory variables produces a model with four independent variables (resampled modern diversity estimate, stepwise time, percentage of shallow occurrences, and latitudinal range) and is referred to henceforth as the "four-variable model" (Fig. 4; Table 1). This model offers a large improvement in predictive power for the value of fossil SQS ( $R^2 = 0.671$ , p < 0.001) and a relatively smaller improvement in the correlation with SQS first differences ( $R^2 = 0.377$ , p < 0.001). Note that none of the temporal variation exhibited by the four-variable model comes from actual change in biodiversity with time, aside from the step increase at the Mesozoic/Cenozoic boundary.

 Table 1. Results of linear correlations between resampled Ocean Biodiversity Information System (OBIS) diversity, recovered fossil diversity, and the additional sample descriptors derived from the Paleobiology Database (PBDB) dataset. AIC, Akaike information criterion. Bold indicates significance at the 0.05 level.

		Raw diversity (SIB)						Sampling standardized (SQS)					
	Value			First differences			Value			First differences			
Single-variable models	Adj. R <sup>2</sup>	p-value	AIC	Adj. R <sup>2</sup>	p-value	AIC	Adj. R <sup>2</sup>	p-value	AIC	Adj. R <sup>2</sup>	p-value	AIC	
Null models													
OBIS	0.408	<0.001	-46.85	0.305	<0.001	-64.64	0.247	<0.001	-34.47	0.184	0.001	-43.29	
OBIS-randomized	0.398	<0.001	-46.03	0.220	<0.001	-59.14	0.204	0.001	-31.75	0.023	0.156	-34.61	
OBIS-brachiopods	0.573	<0.001	_	0.204	0.001	_	0.390	<0.001	_	0.054	0.062	_	
Sampling intensity													
Number of occurrences	0.642	<0.001	-71.08	0.647	<0.001	-97.24	0.183	0.001	-30.54	0.018	0.180	-34.37	
Number of publications	0.760	<0.001	-90.30	0.792	<0.001	-122.63	0.371	<0.001	-43.06	0.206	0.001	-44.57	
% Aragonite occurrences	-0.010	0.463	-21.25	0.006	0.261	-47.51	-0.014	0.549	-20.17	0.019	0.740	-32.60	
Geographic variables													
Number of cells	0.714	<0.001	-81.74	0.706	<0.001	-105.92	0.419	<0.001	-46.89	0.273	<0.001	-48.85	
MST distance	0.699	<0.001	-79.37	0.630	<0.001	-94.96	0.417	<0.001	-46.71	0.204	0.001	-44.46	
Lat. range of occurrences	0.109	0.013	-27.23	0.226	<0.001	-59.53	0.100	0.016	-25.90	0.249	<0.001	-47.24	
Environmental variables													
% Shallow occurrences	0.072	0.036	-23.94	0.056	0.058	-46.90	0.072	0.036	-26.75	0.074	0.035	-37.00	
% Tropical occurrences	0.004	0.282	-21.90	0.005	0.268	-47.47	0.024	0.150	-21.98	0.039	0.095	-35.43	
% Reefal occurrences	-0.021	0.906	-20.69	-0.018	0.676	46.36	-0.019	0.735	-19.91	0.019	0.176	-34.42	
% Carbonate occurrences	0.165	0.002	-30.34	0.007	0.253	-47.55	0.147	0.004	-28.47	-0.006	0.399	-33.23	
Multivariable models													
Mixed models													
Four-variable model	0.754	<0.001	-86.35	0.382	<0.001	-68.44	0.671	<0.001	-71.43	0.377	<0.001	-54.36	
(OBIS + Cenozoic + % shallow + lat. range)													
Best combination (no OBIS)	0.897	<0.001	-128.57	0.827	<0.001	-131.06	0.638	<0.001	-67.64	0.313	<0.001	-50.73	
(no. occurrences + no. cells + % shallow + Cenozoic)													
Groups													
Sampling intensity	0.788	<0.001	-94.32	0.829	<0.001	-129.98	0.410	<0.001	-44.31	0.268	0.001	-46.62	
Geographic variables	0.765	<0.001	-88.54	0.697	<0.001	-102.63	0.407	<0.001	-44.06	0.335	<0.001	-51.25	
Environment	0.157	0.022	-26.00	-0.038	0.683	-42.73	0.159	0.021	-27.12	0.029	0.269	-34.30	
Time models													
Cenozoic step change	0.672	<0.001	-74.57	_	_	_	0.528	<0.001	-57.87	_	_	_	
Continuous increase	0.606	<0.001	-63.38	_	_	_	0.428	<0.001	-47.18	_	_	_	

#### Discussion

Our results indicate that the SQS-corrected view of skeletonized, marine invertebrate, fossil diversity can be reasonably reproduced with a simple null model of geographically resampled modern biodiversity, a binary age parameter, and two dataset variables relating to the geography and environment of sampling. Our null model can be generated without a single fossil taxon ever being named or described, and yet it approximates the very best estimates of both the overall trend of and bin-to-bin changes in Phanerozoic diversity inferred from the fossil record. These results question the validity of all reconstructions of biodiversity trends drawn from the fossil record, even after sampling standardization. Aside from the stepwise increase into the Cenozoic, they are also consistent with the hypothesis that the global number of skeletonized marine invertebrate genera is mostly static (Alroy et al. 2001), with apparent variation resulting almost entirely from changes in the spatial distribution of sampling.

The provocative nature of this statement demands rigorous interrogation of the analysis. Is it possible that our results could be determined solely by inadequacies in the OBIS dataset rather than real variation in the spatial coverage of fossil occurrence data? Do real diversity perturbations like mass extinctions stand out as outliers in the comparison between modeled and fossil record diversity, as they should so long as richness was suppressed for a substantial portion of a time bin? Finally, if our results are legitimate and truly indicate that recovered fossil biodiversity is driven by spatial sampling factors, what is the relative importance of numerical sampling intensity, spatial sampling intensity, and environmental factors?

#### Is OBIS Sufficient?

First, any attempt to describe modern biodiversity, including the OBIS database, is plagued by similar sampling biases, as in the fossil record (Hughes et al. 2021). Inevitably, some regions will be undersampled (e.g., the Southern Ocean; Bonnet-Lebrun et al. 2023), some taxa will be underrepresented (e.g., bryozoans; Kopperud et al. 2022), and some environments will be overrepresented (e.g., shallow coastal habitats; Klein et al. 2019; Shaw et al. 2020). Klein et al. (2019) note that half of described marine species are not found in OBIS (see also Webb and Vanhoorne 2020), and those that are tend to be common and well known. However, a closer look at the OBIS dataset in comparison with PBDB data shows that our results are unlikely to arise from spatial or other biases within OBIS. For example, while more records come from Europe and North America in the OBIS, mirroring the socioeconomic and access issues seen in the PBDB (Fig. 5A), the bias is much less pervasive or impactful. Even where raw sampling numbers are relatively lower, biodiversity hotspots in places like the Indo-Pacific, Gulf of Mexico, and the African south coast still stand out within our subset of the OBIS dataset (Fig. 5B). By comparing data from the most recent PBDB time bin (the late Miocene-Quaternary) with the OBIS dataset, it becomes clear that OBIS is better sampled across latitude (Fig. 6) and taxonomy (Fig. 7). Southern Hemisphere sampling in the PBDB is comparatively incomplete, and latitudinal sampling deficiencies are only worse in deep time. The most recent PBDB time bin is also overwhelmingly dominated by mollusks, especially bivalves (69% and 41%, respectively), while the OBIS contains a taxonomically broader suite of organisms (only 37% mollusk and 18% bivalve; see also Webb and Vanhoorne 2020). The median



**Figure 5.** Ocean Biodiversity Information System (OBIS) occurrences and richness in our subset of the data. **A**, OBIS occurrences within equal-area cells. **B**, OBIS generic richness within equal-area cells. Patterns in **A** are largely driven by spatial sampling biases (e.g., a greater number of scientific institutions in the Northern Hemisphere), but important marine invertebrate diversity hotspots can be seen in **B**, implying that imperfect spatial sampling has not greatly distorted spatial diversity patterns in the OBIS.



Figure 6. Latitudinal sampling in the Ocean Biodiversity Information System (OBIS) dataset (blue) versus the most recent (mid-Miocene to Recent) time bin of the Paleobiology Database (PBDB) (red). OBIS sampling is more complete in the Southern Hemisphere (A), but both exhibit the latitudinal diversity gradient (B).



Figure 7. The number of phyla per sampled cell in the Ocean Biodiversity Information System (OBIS) dataset (A) versus the most recent time bin of the Paleobiology Database (PBDB) (B).

OBIS cell contains five of the six phyla examined in this study, while the median recent PBDB cell only contains two. Finally, comparing cells that are occupied by occurrences in both datasets shows that well-represented cells in the PBDB are typically well sampled in the OBIS, but the reverse is not true (Fig. 8). In sum, there are many more cells sampled in the OBIS *and* those cells are consistently better sampled than those of the PBDB.



**Figure 8.** The numbers of families present in cells occupied in both the Ocean Biodiversity Information System (OBIS) and most recent Paleobiology Database (PBDB) data. All diverse PBDB cells are also diverse in OBIS, but the reverse is not the case.

Even if sampling of the OBIS is not complete—biological sampling can never be considered 100% comprehensive—sampling within the OBIS is certainly comprehensive enough for comparisons to the PBDB and for global spatial diversity statistics.

Second, our results cannot be dismissed as noise. If spatial sampling factors are important for driving the similarity between SQS diversity and resampled modern diversity, then removing the spatial component of the dataset through randomization should eliminate the correlation, and this is exactly what happens. When the taxonomic column of the OBIS dataset is randomized, effectively removing all relationships between space and the distribution of organisms on the Earth, the model fails to accurately predict bin-to-bin change in the SQS value of the fossil record (adj.  $R^2 = 0.023$ , p = 0.156; Table 1). Furthermore, all versions of the model, including the spatially randomized one, predict bin-to-bin change in SQS value better than does the number of fossil occurrences through time. If SQS reduced the influence of spatial sampling bias as much as it does numerical sampling bias, we would expect the correlation between SQS and the number of occurrences through time to be similar to the correlation between SQS and resampled modern diversity, and this is not the case (Table 1).

Finally, our model is methodologically complex but theoretically simple. From a theoretical standpoint, the model can be conceived as sampling the modern landscape of biodiversity with the same or similar spatial fidelity as that in the fossil record. If the spatial distribution of fossil sampling were sufficient, there should be no appreciable change in predicted richness, as the same data are being sampled each time. There is no particularly good reason to believe that marine biodiversity in deep time was spatially arrayed in a similar manner to modern biodiversity-in fact there are many reasons to believe this not to be the case. While some manifestation of the LDG seems to be more or less persistent, the shape of that relationship has varied significantly over time, from unimodal to bimodal to flat (Crame 2001; Alroy et al. 2008; Mannion 2020), and the latitudinal and longitudinal distribution of taxa have clearly changed with the changing configuration of continents and ocean basins (Valentine 1973). Nevertheless, the modern landscape can suffice as a null hypothesis for our spatial sampling model, and if the spatial distribution of organisms on the Earth has changed dramatically through deep time, it should be relatively easy to reject our null hypothesis. Again this is not the case. Our results suggest that, when estimating trends or changes in global-scale diversity, even SQS-standardized fossil diversity remains substantially impacted by spatial sampling factors.

#### Insights from Model Residuals

If the resampled OBIS model can be considered a null model of expected SQS fossil diversity, a closer look at time bins when the raw model and subsequent adjustments perform poorly may be informative.

First, a fundamental question arising from paleobiodiversity studies is whether the dramatic apparent increase in biodiversity into the Cenozoic and toward the present is real or an artifact (Alroy et al. 2001). Consistent with Alroy (2010), our analysis of SQS richness shows an approximately twofold increase in the average richness of Cenozoic time bins compared with those of the Paleo-Mesozoic, even when unconsolidated sediments are excluded from the analysis. To test the hypothesis of a step change in marine biodiversity following the Cretaceous–Paleogene mass extinction, we compare models that include no time-dependent variable, a "Cenozoic or not" factor, and time as a continuous variable (i.e., bin age in millions of years). Results (Table 1) clearly favor the "Cenozoic or not" model as the model with the most explanatory power for both SIB and SQS fossil diversity estimates, indicating the likelihood of a real stepwise increase in marine biodiversity starting near the Cretaceous/Paleogene boundary. However, geographic sampling proxies like the resampled OBIS diversity estimate and the summed MST length of PBDB-sampled cells accurately explain the observed increase in SQS and SIB richness during the Cenozoic, indicating little support for a continuous increase in richness among marine invertebrates over the last 66 Myr. Close and colleagues likewise found a stepwise increase in the biodiversity of marine animals (Close et al. 2020b) and terrestrial vertebrates (Close et al. 2020a) in the early Cenozoic that they could not attribute to a spatial sampling artifact. Whether this observed stepwise increase is real or the result of additional bias (e.g., more splitting of Cenozoic genera; Hendy 2010) remains an open question and deserves closer investigation.

After correcting for the step up into the Cenozoic, residuals correlate weakly but significantly with two variables drawn from the PBDB dataset (Fig. 4A). First, the percentage of fossil occurrences from relatively shallow environments (roughly, shallower than benthic assemblage zone 4, exclusive) correlates positively with the residuals (adj.  $R^2 = 0.193$ , p = 0.001). Residual correlation with an environmental variable is unsurprising, because the OBIS model does not take any paleoenvironmental information into account, and biological diversity certainly varies across environments. However, to the degree that times with proportionately more fossils from shallow environments also reflect times of greater continental flooding and expansion of shallow seas, this correlation could reflect a real signal of biotic response to expansion and contraction of habitat area (Peters and Husson 2017). The extent of shallow-marine sediment is also seen to correlate positively with spatially subsampled diversity by Close et al. (2020b), suggesting species-area effects (Rosenzweig 1995) not predicted by sampling alone. The significant middle Cambrian overestimation of PBDB diversity, for example, disappears in the four-variable OBIS model that incorporates percent shallow occurrences (Figs. 4A, 9), perhaps reflecting a real increase in diversity during the expansion of shallow-water habitats associated with continental flooding during the Sauk sequence. Given the nature of Paleo-Mesozoic fossil data, a relative increase in the sampling of inferred deep occurrences (or a reduction in the proportion of shallow occurrences) might additionally reflect times of increased ocean anoxia. Deep environments are often inferred from the presence of black shales and dark-colored micrites, which are typically depauperate and more directly indicative of anoxia than they are of deep water (Smith et al. 2019). Anoxia might therefore contribute to the forces producing relatively lower than expected fossil SQS diversity (e.g., Peters 2007) in intervals with proportionally less shallow sampling.

Several other environmental variables were tested, including the percentage of tropical and reefal occurrences, both of which are often associated with higher origination and standing diversity in the literature (Aberhan and Kiessling 2012), but none explained any significant proportion of variation in the residuals. Rather than showing that these variables are not important for standing diversity, this indicates that modern and paleoenvironmental trends diverge in this respect—in the modern dataset, the tropics and reefal locations have not changed from where they were in the past, but the location of the shallow ocean certainly has. In other words, the modern dataset already has a spatial concentration of diversity at the tropics (and at reefs, which are mostly tropical), so no additional information from the fossil dataset is necessary to explain these spatial patterns. The only other variable to significantly correlate with model residuals was a simple measure of the absolute latitudinal range of fossil occurrences in each time bin (adj.  $R^2 = 0.098$ , p = 0.017), an unsurprisingly weak correlation, given that the model already attempts to account for latitudinal variation in sampling. Its incorporation as an additional explanatory variable implies that the original OBIS model may have treated latitude with too little granularity.

The OBIS Cenozoic-corrected and four-variable models tend to overestimate richness in time bins at or immediately after the major mass extinctions (Fig. 4A), but model residuals for most of these time bins (Fig. 9) do not stand out, reinforcing that rebound from extinctions largely happens within the succeeding time bin at this scale, as suggested by other studies of regional and global biodiversity (Alroy et al. 2008; Close et al. 2020b). The Late Devonian biodiversity crisis is an exception, perhaps owing to its duration over a longer interval of time. PBDB richness is much lower compared with the OBIS model predictions at the Jurassic/Cretaceous (J/K) boundary (Fig. 4A). The latter is especially interesting given that the J/K boundary is generally poorly understood and may be an interval especially affected by sampling biases (Tennant et al. 2016). On the other hand, the interval from the end-Silurian to the Middle Devonian stands out as a period with uncharacteristically high residuals-that is, fossil diversity is higher than expected by the OBIS model. Close et al. (2020b) also find SQS diversity to be elevated at this time relative to their spatially standardized analysis. One might infer this to reflect an expansion in marine biodiversity (Klug et al. 2010; Ruban 2013), yet the interval encompassing the well-known Ordovician radiation does not stand out as a time of elevated richness compared with model estimates. The Early-Middle Devonian, however, is a particularly spatially well-sampled interval, specifically with an atypically high latitudinal range of sampling, even though the number of occurrences collected from that interval is unremarkable.



**Figure 9.** Residuals between the resampled four-variable Ocean Biodiversity Information System (OBIS) model and fossil shareholder quorum subsampling (SQS) diversity. Dashed lines show the ages of the "big five" mass extinctions. CM, Cambrian; O, Ordovician; S, Silurian; D, Devonian; C, Carboniferous; P, Permian; Tr, Triassic; J, Jurassic; K, Cretaceous; Pg, Paleogene; Ng, Neogene.

#### Moving Forward

What can be done to remedy or address the effects of spatial bias in the fossil record? One line of thought is to simply increase sampling efforts in previously undersampled locations, including across the Global South. Given that the density of sampling is much higher in North America and Europe than in the rest of the world (Ye and Peters 2023), much could be learned from this process, and many gaps in knowledge might be filled. However, it is likely that biases originating from the spatially nonrandom distribution of exposed sedimentary rock of appropriate age (rather than from sociopolitical phenomena) would still influence interpretations of the fossil record, even after substantial sampling efforts, because the samples necessary to fill the gaps simply do not exist. Another approach is to focus on numerically standardized paleobiological diversity within multiple areas of comparable spatial extent drawn from each time interval (Close et al. 2020a,b), allowing for direct comparisons across geological time at spatially consistent scales. While both increased sampling and improved understanding of diversity variance within and across equal spatial regions are important for addressing the spatial structure inherent in the fossil record, spatially informed quantitative and statistical methods are critical as well. Paleobiologists will benefit from new focus on the development of novel spatial resampling techniques (e.g., recent work by Antell et al. [2024]) and from interdisciplinary studies with ecologists and neontologists experienced in dealing with spatial sampling questions.

Additionally, future efforts to address spatial bias in the sampling-standardized record of fossil diversity should focus on differentiating the real biogeographic effects imposed by plate tectonics, continental flooding, and global climate change from the artificial variation introduced by the distribution of sampling. The spatial distribution of marine taxa today is not likely to approximate that in, for example, the Middle Devonian, because in the latter, the Northern Hemisphere was mostly devoid of land or shallow water (Scotese 2016) and the continents were less dispersed than today (Zaffos et al. 2017). Likewise for the Cretaceous hothouse, with much warmer temperatures and no ice at the poles. As well, given the preponderance of data from North America and Europe and their northward shift out of the tropics over the last ~200 Myr, if anything like the modern LDG is typical of the Phanerozoic, numerically standardized estimates alone are still likely to underestimate mid-late Mesozoic diversity in comparison to the Paleozoic. To what degree are recovered patterns in paleobiodiversity in space and time real and due to such factors, as opposed to driven purely by spatial (and numerical) sampling limitations?

In conclusion, we show that a subset of the OBIS dataset of modern global biodiversity with taxonomic membership comparable to the skeletonized marine invertebrate fossil record, when sampled spatially as dictated by the fossil record, produces predicted richness values and first differences that reasonably correlate with fossil richness estimates, even after state-of-the-art numerical subsampling procedures have been applied. As also deduced by Close et al. (2020b) using entirely different methods, these results strongly caution the overinterpretation of biodiversity trends inferred from the fossil record and are consistent with the argument that global marine biodiversity has been largely static over the Phanerozoic, with apparent variation resulting almost entirely from changes in the spatial distribution of sampling. Changes in spatial sampling parameters, like the number and distribution of geographic cells sampled, their summed MST distances, and their latitudinal range, are at least as important for predicting fossil richness values as factors associated with numerical sampling intensity. This analysis highlights the importance of the spatial distribution of sampling in analyses of large-scale fossil biodiversity and echo the call from Antell et al. (2024) for the development of new methodologies that explicitly account for spatial biases in the fossil record.

Acknowledgments. We thank all of the original authors, data enterers, programmers, and contributors whose work produced the Paleobiology Database and the Ocean Biodiversity Information System. Without you, these kinds of analyses would not be possible. M. Clapham and an anonymous reviewer provided valuable feedback that helped refine the article, and we are grateful to them. Special thanks to J. Kastigar in the editorial office for patience and understanding. DBP was supported by Collaboration for Unprecedented Success and Excellence (CUSE) grant II-38-2020 from Syracuse University's Office of Research to LCI and fellowships from the Department of Earth Sciences and the Graduate School. This paper is Publication No. 493 for the Paleobiology Database.

Competing Interests. The authors declare no competing interests.

**Data Availability Statement.** Data and R code are available from the Dryad Digital Repository: https://doi.org/10.5061/dryad.2280gb621.

#### **Literature Cited**

- Aberhan, M. 1992. Palokologie und zeitliche Verbreitung benthischer Faunengemeinschaften im Unterjura von Chile. *Beringeria* 5:1-174.
- Aberhan, M., and W. Kiessling. 2012. Phanerozoic marine biodiversity: a fresh look at data, methods, patterns and processes. Pp. 3–22 in J. A. Talent, ed. Earth and life: global biodiversity, extinction intervals and biogeographic perturbations through time. Springer, New York.
- Alroy, J. 2010. Geographical, environmental and intrinsic biotic controls on Phanerozoic marine diversification. *Palaeontology* 53:1211–1235.
- Alroy, J., C. R. Marshall, R. K. Bambach, K. Bezusko, M. Foote, F. T. Fürsich, T. A. Hansen, S. M. Holland, L. C. Ivany, D. Jablonski, D. K. Jacobs, D. C. Jones, M. A. Kosnik, S. Lidgard, S. Low, A. I. Miller, P. M. Novack-Gottshall, T. D. Olszewski, M. E. Patzkowsky, D. M. Raup, K. Roy, J. J. Sepkoski, Jr., M. G. Sommers, P. J. Wagner, and A. Webber. 2001. Effects of sampling standardization on estimates of Phanerozoic marine diversification. *Proceedings of the National Academy of Sciences USA* 98:6261–6266.
- Alroy, J., M. Aberhan, D. J. Bottjer, M. Foote, F. T. Fursich, P. J. Harries, A. J. W. Hendy, S. M. Holland, L. C. Ivany, W. Kiessling, M. A. Kosnik, C. R. Marshall, A. J. McGowan, A. I. Miller, T. D. Olszewski, M. E. Patzkowsky, S. E. Peters, L. Villier, P. J. Wagner, N. Bonuso, P. S. Borkow, B. Brenneis, M. E. Clapham, L. M. Fall, C. A. Ferguson, V. L. Hanson, A. Z. Krug, K. M. Layou, E. H. Leckey, S. Nürnberg, C. M. Powers, J. A. Sessa, C. Simpson, A. Tomasovych, and C. C. Visaggi. 2008. Phanerozoic trends in the global diversity of marine invertebrates. *Science* 321:97-100.
- Antell, G. T., R. B. Benson, and E. E. Saupe. 2024. Spatial standardization of taxon occurrence data—a call to action. *Paleobiology* 50:177–193.
- Barnes, R., and K. Sahr. 2017. dggridR: Discrete Global Grids for R. https:// cran.r-project.org/web/packages/dggridR/index.html. Accessed June 2022.
- Bonnet-Lebrun, A.-S., M. Sweetlove, H. J. Griffiths, M. Sumner, P. Provoost, B. Raymond, Y. Ropert-Coudert, and A. P. Van de Putte. 2023. Opportunities and limitations of large open biodiversity occurrence databases in the context of a marine ecosystem assessment of the Southern Ocean. Frontiers in Marine Science 10:1150603.
- Bozdogan, H. 1987. Model selection and Akaike's information criterion (AIC): the general theory and its analytical extensions. *Psychometrika* 52:345–370.
- Buffon, G. L. Leclerc, comte de. 1774. Histoire naturelle, générale et particuliére: De la manière d'étudier & de traiter l'histoire naturelle. Histoire & théorie de la terre. L'Imprimerie Royale, 1750 Vol. 1.

- Chao, A., and L. Jost. 2012. Coverage-based rarefaction and extrapolation: standardizing samples by completeness rather than size. *Ecology* 93:2533–2547.
- Close, R. A., S. W. Evers, J. Alroy, and R. J. Butler. 2018. How should we estimate diversity in the fossil record? Testing richness estimators using sampling-standardised discovery curves. *Methods in Ecology and Evolution* 9:1386–1400.
- Close, R. A., R. B. J. Benson, J. Alroy, M. T. Carrano, T. J. Cleary, E. M. Dunne, P. D. Mannion, M. D. Uhen, and R. J. Butler. 2020a. The apparent exponential radiation of Phanerozoic land vertebrates is an artefact of spatial sampling biases. *Proceedings of the Royal Society B* 287:20200372.
- Close, R. A., R. B. J. Benson, E. E. Saupe, M. E. Clapham, and R. J. Butler. 2020b. The spatial structure of Phanerozoic marine animal diversity. *Science* **368**:420–424.
- Cooper G. A. and Grant, R. E. 1977. Permian Brachiopods of West Texas, VI. Smithsonian Contributions to Paleobiology 32:3161–3370.
- Crame, J. A. 2001. Taxonomic diversity gradients through geological time. Diversity and Distributions 7:175–189.
- Crampton, J. S., A. G. Beu, R. A. Cooper, C. M. Jones, B. Marshall, and P. A. Maxwell. 2003. Estimating the rock volume bias in paleobiodiversity studies. *Science* 301:358–360.
- Csardi, G., and T. Nepusz. 2006. The igraph software package for complex network research. *InterJournal Complex Syst* 1695:1–9. https://igraph.org. Last accessed June 2022.
- Dallas, T. A., L. A. Holian, and C. T. Caten. 2024. Geographic and temporal distance–decay relationships across taxa. Oikos 2024:e10269.
- Foster, W. J., C. L. Garvie, A. M. Weiss, A. D. Muscente, M. Aberhan, J. W. Counts, and R. C. Martindale. 2020. Resilience of marine invertebrate communities during the early Cenozoic hyperthermals. *Scientific Reports* 10:2176.
   Fürsich, F. T. 1999. Unpublished data. Paleobiology Database Ref. 41.
- He, S., C. Qin, and A. J. Soininen. 2024. A flexible framework to assess pat-
- terns and drivers of beta diversity across spatial scales. *Ecography* 2024: e06901.
- Hendy, A. J. W. 2010. Taphonomic overprints on Phanerozoic trends in biodiversity: lithification and other secular megabiases. Pp. 19–77 in P. A. Allison and D. J. Bottjer, eds. *Taphonomy: process and bias through time*. Springer, New York.
- Holland, S. M., and M. E. Patzkowsky. 2007. Gradient ecology of a biotic invasion: biofacies of the type Cincinnatian Series (Upper Ordovician), Cincinnati, Ohio region, USA. *Palaios* 22:408–423.
- Hughes, A. C., M. C. Orr, K. Ma, M. J. Costello, J. Waller, P. Provoost, Q. Yang, C. Zhu, and H. Qiao. 2021. Sampling biases shape our view of the natural world. *Ecography* 44:1259–1269.
- Humboldt, A. von, and A. Bonpland. 1807. Ideen zu einer Geographie der Pflanzen nebst einem Naturgemälde der Tropenländer: auf Beobachtungen und Messungen gegründet, welche vom 10ten Grade nördlicher bis zum 10ten Grade südlicher Breite, in den Jahren 1799, 1800, 1801, 1802 und 1803 angestellt worden sind. Bey F.G. Cotta, Tübingen.
- Jones, L. A., C. D. Dean, P. D. Mannion, A. Farnsworth, and P. A. Allison. 2021. Spatial sampling heterogeneity limits the detectability of deep time latitudinal biodiversity gradients. *Proceedings of the Royal Society B* 288: p. 20202762.
- Jung, P. 1989. Revision of the Strombina-group (Gastropoda; Columbellidae), fossil and living. Schweierische Paläontologische Abhandlungen 111:1–298.
- King, R. E. 1931. The Geology of the Glass Mountains, Texas, Part II, Faunal summary and correlation of the Permian formations with description of Brachiopoda. University of Texas Bulletin 3042:1–245.
- Klein, E., W. Appeltans, P. Provoost, H. Saeedi, A. Benson, L. Bajona, A. C. Peralta, and R. S. Bristol. 2019. OBIS infrastructure, lessons learned, and vision for the future. *Frontiers in Marine Science* 6:588.
- Klug, C., B. Kröger, W. Kiessling, G. L. Mullins, T. Servais, J. Frýda, D. Korn, and S. Turner. 2010. The Devonian nekton revolution. *Lethaia* 43:465–477.
- Kocsis, Á. T., C. J. Reddin, J. Alroy, and W. Kiessling. 2019. The R package divDyn for quantifying diversity dynamics using fossil sampling data. *Methods in Ecology and Evolution* 10:735–743.
- Kopperud, B. T., S. Lidgard, and L. H. Liow. 2022. Enhancing georeferenced biodiversity inventories: automated information extraction from literature records reveal the gaps. *PeerJ* 10:e13921.

- Manivit, J., Y.-M. Le Nindre, and D. Vaslet. 1990. Le Jurassique d'Arabie Centrale. Document du BRGM 4(194):25–519.
- Mannion, P. D. 2020. A deep-time perspective on the latitudinal diversity gradient. Proceedings of the National Academy of Sciences USA 117:17479–17481.
- McGowan, A. J., and A. B. Smith. 2008. Are global Phanerozoic marine diversity curves truly global? A study of the relationship between regional rock records and global Phanerozoic marine diversity. *Paleobiology* 34:80–103.
- Patzkowsky, M. E., and S. M., Holland. 2012. Stratigraphic paleobiology: understanding the distribution of fossil taxa in time and space. University of Chicago Press, Chicago.
- Pebesma, E., R. Bivand, E. Racine, M. Sumner, I. Cook, T. Keitt, R. Lovelace, et al. 2018. Simple features for R: Standardized support for spatial vector data. R Journal 10:439–446.
- Peters, S. E. 2007. The problem with the Paleozoic. Paleobiology 33:165-181.
- Peters, S. E., and J. M. Husson. 2017. Sediment cycling on continental and oceanic crust. *Geology* 45:323–326.
- Phillips, J. 1860. Life on the Earth: its origin and succession. Macmillan and Company, Cambridge.
- Provoost, P., S. Bosch, and W. Appeltans. 2022. robis: Ocean Biodiversity Information System (OBIS) client. https://CRAN.R-project.org/package=robis. Run on June 7, 2022.
- Raja, N. B., E. M. Dunne, A. Matiwane, T. M. Khan, P. S. Nätscher, A. M. Ghilardi, and D. Chattopadhyay. 2022. Colonial history and global economics distort our understanding of deep-time biodiversity. *Nature Ecology* and Evolution 6:145–154.
- Raup, D. M. 1976. Species diversity in the Phanerozoic: an interpretation. Paleobiology 2:289–297.
- Reboulet, S. 1996. L'évolution des ammonites du Valanginien-Hauterivien inférieur du bassin Vocontien et de la plate-form Provençale (sud-est de la France): relations avec la stratigraphie séquentielle et implications biostratigraphiques. Documents des Laboratoires de Géologie Lyon 137:1–371.
- Reed, F. R. C. 1944. Brachiopoda and Mollusca from the Productus limestones of the Salt Range. Palaeontogica Indica, Memoirs of the Geological Survey of India, New Series 23(2):1–596.
- Rosenzweig, M. L. 1995. Species diversity in space and time. Cambridge University Press, Cambridge.
- Ruban, D. 2013. Was there more space in the late Early Devonian for marine biodiversity to peak than in the early Late Ordovician? A brief note. *Geoloski anali Balkanskoga poluostrva* 2013(74):1–8.
- Schopf, T. J. 1978. Fossilization potential of an intertidal fauna: Friday Harbor, Washington. *Paleobiology* 4:61–270.
- Scotese, C. R. 2016. PALEOMAP PaleoAtlas for GPlates and the PaleoData Plotter Program, PALEOMAP Project. https://www.earthbyte.org/ paleomap-paleoatlas-for-gplates/. Accessed June 1, 2022.
- Shaw, J. O., D. E. Briggs, and P. M. Hull. 2020. Fossilization potential of marine assemblages and environments. *Geology* 49:258–62.
- Smith, A. B., and A. J. M. Gowan. 2007. The shape of the Phanerozoic marine palaeodiversity curve: how much can be predicted from the sedimentary rock record of Western Europe? *Paleobiology* 50:765–774.
- Smith, L. B., J. Schieber, and R. Wilson. 2019. Shallow-water onlap model for the deposition of Devonian black shales in New York, USA. *Geology* 48:279–283.
- Tennant, J. P., P. D. Mannion, and P. Upchurch. 2016. Sea level regulated tetrapod diversity dynamics through the Jurassic/Cretaceous interval. *Nature Communications* 7:12737.
- Valentine, J. W. 1973. Evolutionary paleoecology of the marine biosphere. Prentice-Hall, Englewood Cliffs, N.J.
- Wall, P. D., L. C. Ivany, and B. H. Wilkinson. 2009. Revisiting Raup: exploring the influence of outcrop area on diversity in light of modern sample-standardization techniques. *Paleobiology* 35:146–167.
- Webb, T. J., and B. Vanhoorne. 2020. Linking dimensions of data on global marine animal diversity. *Philosophical Transactions of the Royal Society B* 375:20190445.
- Ye, S., and S. E. Peters. 2023. Bedrock geological map predictions for Phanerozoic fossil occurrences. *Paleobiology* 49:1–20.
- Zaffos, A., S. Finnegan, and S. E. Peters. 2017. Plate tectonic regulation of global marine animal diversity. *Proceedings of the National Academy of Sciences USA* 114:5653–5658.